

# DETERMINANTS OF THE VALUE OF GREEN BONDS IN THE UNITED KINGDOM

Tran Thi Nhi Sheffield Business School Sheffield Hallam University United Kingdom

Submitted 3oth September 2024 Accepted 7<sup>th</sup> March, 2025 Published 7<sup>th</sup> March, 2025

#### **ABSTRACT**

This dissertation investigates the determinants influencing the market value of green bonds in the United Kingdom, focusing on a range of factors including credit ratings, environmental impact scores, regulatory frameworks, market conditions, and liquidity. The research employs quantitative methodologies to analyze data collected from various financial databases and green bond reports, offering a comprehensive examination of the unique dynamics that affect green bond pricing within the UK's robust financial market.

The study identifies significant relationships between the market value of green bonds and their associated credit ratings, the perceived environmental benefits of funded projects, as well as the extent of regulatory support. Additionally, it explores how broader economic conditions, and the liquidity of green bonds impact their yield spreads compared to conventional bonds. The findings suggest that while credit ratings and regulatory support are critical, the environmental impact scores significantly enhance the appeal and market value of green bonds, aligning financial gains with sustainable investment outcomes.

This dissertation contributes to the existing literature by providing empirical evidence on the nuanced factors that influence green bond values, addressing a significant gap in sustainable finance research. The insights gained from this study are intended to assist investors, policymakers, and financial institutions in optimizing green bond frameworks to better meet environmental targets and investor expectations in the UK and potentially, on a global scale

FinTech and Artificial Intelligence in Finance (FINTAF)
Issue 2

FinTech and Artificial Intelligence in Finance (FINTAF)
Issue 2

#### INTRODUCTION

## 1.1 Research background

The increasing urgency to combat climate change has catalyzed the evolution of financial instruments aimed at promoting sustainable economic growth. Among these, green bonds have emerged as a crucial vehicle for funding environmentally friendly projects, with significant growth observed globally and within the United Kingdom (Tolliver, 2020). Initially introduced in the early 2000s, green bonds are designed to raise capital for projects that offer environmental benefits such as renewable energy, biodiversity conservation, and climate change mitigation (Sartzetakis, 2020).

The UK has become a focal point for green finance, driven by its commitment to reduce carbon emissions and bolster sustainability initiatives. The government's pledge to bring all greenhouse gas emissions to net zero by 2050 necessitates expansive investment in green projects, positioning green bonds as a key element of this financial strategy (Vella, 2021). This aligns with the global financial community's shift towards investments that not only yield financial returns but also contribute positively to environmental sustainability (Hannon, 2023).

However, despite the market's rapid growth, the factors influencing the value of green bonds remain complex and multifaceted (Mertzanis & Tebourbi, 2024). They include traditional financial metrics like credit risk and yield, alongside newer considerations such as the environmental impact score of the projects they fund and the evolving regulatory landscape. Deschryver and Demariz (2020) recommended that understanding these determinants is

essential not only for investors seeking profitable and ethical investments but also for policymakers aiming to craft regulations that effectively stimulate the green bond market.

In the UK, the green bond market's dynamics are influenced by both global economic trends and local market conditions. This includes the Bank of England's monetary policies, the financial health of issuers, and public sentiment towards green initiatives, all of which play a role in shaping the market's response to green bonds (Bank of England, 2021). According to Nykvist and Maltais (2022), the growth of the market is also indicative of broader shifts in the financial sector towards more sustainable investment practices, reflecting a changing investor base that increasingly values environmental impact in its portfolio decisions.

This dissertation explores these dynamics within the UK context, providing an analysis that not only enhances understanding of green bond valuation but also contributes to the broader discourse on sustainable finance. By examining the various determinants of green bond value, this research aims to uncover insights that can help optimize the framework within which green bonds operate, thereby supporting the UK's environmental and financial objectives.

### 1.2 Problem Statement

As the global financial sector increasingly commits to addressing environmental challenges, green bonds have emerged as vital instruments for fostering sustainable growth (Menon &, 2022). However, Alamgir and Cheng (2023) stated that despite their growing popularity and critical role in funding sustainable projects, the determinants of the value of green bonds in the United Kingdom remain insufficiently understood. This lack of clarity presents a significant problem for investors, issuers, and policymakers who seek to maximize the effectiveness and attractiveness of these financial instruments.

In the UK, while green bonds have gained momentum, underpinned by strong governmental support and a proactive financial market, the precise factors influencing their pricing and investment attractiveness are not fully elucidated (Robins & Gouldson, 2019). Critical aspects such as the impact of credit ratings, the perceived environmental effectiveness of funded projects, and the regulatory environment are assumed to influence green bond values, but the extent and nature of these influences are not empirically quantified in the current body of literature. Moreover, market conditions such as liquidity and economic stability, which play significant roles in the valuation of traditional bonds, are often overlooked in the context of green bonds (Bhutta, 2022).

The problem is compounded by the 'greenium' - a premium that green bonds can command over conventional bonds—which varies significantly and unpredictively. Understanding what drives this premium, or in some cases, why it fails to appear, is essential for the maturation of the green bond market (Ando, 2023). This inconsistency in greenium not only hinders the predictability of investment returns but also raises questions about the efficacy of green bonds as tools for environmental impact financing (Macaskill, 2020).

Furthermore, there is a critical gap in research that connects macroeconomic factors with the performance of green bonds specifically in the UK. As the UK aims to lead in green finance, detailed knowledge of how broader economic conditions affect green bond values is crucial for formulating effective financial and environmental policies (Mankata, 2020).

This dissertation seeks to address these gaps by conducting a comprehensive analysis of the determinants of green bond values in the UK. By identifying and quantifying the impact of various factors on green bond pricing, this study aims to enhance the understanding and strategic use of green bonds, contributing to more informed decision-making and effective policy formulations in sustainable finance.

### 1.3 Research Aims and Objectives

The aim of this dissertation is to investigate the determinants of the market value of green bonds in the United Kingdom. This study seeks to identify and dissect the multiple dimensions—economic, environmental, and regulatory—that potentially drive investor decisions and market performance of green bonds. The justification for focusing on the UK stems from its pioneering role in the global green finance sector, coupled with the recent governmental pushes towards sustainable investment practices as part of its green finance strategy. Understanding these dynamics in the UK context not only provides insights into the effectiveness of current policies but also informs future regulatory and market developments.

## **Research Objectives:**

To achieve the overarching aim, the research is structured around the following specific objectives:

- To investigate how the credit ratings of issuers affect the yield spreads of green bonds relative to conventional bonds
- To determine the influence of environmental impact scores on the market value of green bonds
- To explore the role of government policies and incentives in shaping the market dynamics of green bonds
- To study the effects of broader macroeconomic conditions and specific market factors, such as interest rates and economic growth, on the pricing of green bonds
- To analyze the significance of liquidity in the trading and valuation of green bonds

These objectives are crafted not only to explore comprehensive dimensions of green bond valuation but also to offer actionable insights for policymakers, investors, and financial

analysts focused on sustainable finance. By investigating these facets, the dissertation intends to provide a robust analysis that contributes significantly to both academic research and practical financial strategies in the realm of green finance. Each objective is designed to collectively build upon the others, ensuring a holistic understanding of the green bond market's current status and its future trajectory.

## 1.4 Research question

This study seeks to explore the key factors influencing the value of green bonds in the UK through the following research questions:

- 1. How does the credit rating of bond issuers influence the yield spreads of green bonds compared to conventional bonds in the UK?
- 2. What impact do environmental impact scores have on the market prices of green bonds?
- 3. How do regulatory frameworks and government incentives impact the pricing and market dynamics of green bonds?
- 4. In what ways do macroeconomic conditions affect the valuation of green bonds in the UK?
- 5. What role does liquidity play in determining the market value of green bonds, and how does it compare to conventional bonds?

## 1.5 Significance of the Research

This research is significant for several reasons. Firstly, it addresses a critical gap in the existing literature by providing empirical analysis on the factors influencing green bond values in the United Kingdom—a market that plays a pivotal role in global sustainable finance. Understanding these determinants is crucial for investors seeking to optimize their

portfolios while contributing to environmental sustainability. Additionally, the findings from this study have the potential to inform policymakers and financial regulators on how to enhance the effectiveness of green bonds as tools for achieving environmental goals. By analyzing key factors such as credit ratings, environmental impact scores, and macroeconomic conditions, this research will contribute to the development of more robust frameworks for green bond valuation and investment strategies, ultimately supporting the UK's commitment to sustainable development.

#### 1.6 Structure of the Dissertation

This dissertation is organized into six key chapters. Chapter 1 introduces the research topic, providing the context, background, problem statement, research aims, objectives, and questions, along with the scope, ethics, and limitations of the study. Chapter 2 offers a comprehensive literature review, critically analyzing existing research on green bonds and identifying gaps that this study aims to fill. Chapter 3 outlines the research methodology, detailing the quantitative approach, data collection, and analysis techniques employed to investigate the determinants of green bond values in the UK. Chapter 4 presents the results of the empirical analysis, while Chapter 5 discusses these findings in the context of the broader literature, highlighting their implications for theory and practice. Finally, Chapter 6 concludes the dissertation, summarizing the key findings, discussing their significance, and offering recommendations for future research and practical applications.

### 1.7 Chapter conclusion

In summary, this first chapter has introduced the primary research focus: the factors influencing green bond values in the United Kingdom. This study is aligned with the UK's emphasis on advancing sustainable finance and achieving its environmental objectives. It seeks to offer a distinctive contribution to the existing literature by investigating key

determinants such as credit ratings, environmental impact scores, regulatory frameworks,

macroeconomic conditions, and liquidity, which are essential for understanding green bond

valuation. By analyzing these factors within the UK's financial market from 2015 to 2023, this

research will deliver comprehensive insights into the drivers of green bond values. The

following chapter will delve into a literature review pertinent to this research.

**CHAPTER 2**:

LITERATURE REVIEW

2.1 Introduction

This chapter undertakes a critical examination of the scholarly literature surrounding green

bonds, focusing specifically on their valuation determinants within the UK market. This review

serves multiple purposes: it contextualizes green bonds within the broader field of sustainable

finance, identifies the economic, environmental, and regulatory factors influencing their

market value, and highlights the research gaps that this dissertation aims to address. The

relevance of this inquiry is underscored by the increasing emphasis on sustainable investment

solutions in addressing global environmental challenges.

2.2 Green Bonds: Definition and Growth

Green bonds are defined as fixed-income securities designed specifically to raise capital for

projects with environmental benefits, such as renewable energy, energy efficiency, sustainable

waste management, and biodiversity conservation (Gilchrist, 2021). The primary distinction

between green bonds and conventional bonds lies in the explicit commitment by the issuer to

use the raised funds for environmentally sustainable projects (Maltais & Nykvist,

2020). The World Bank issued the first labeled green bond in 2008, setting a precedent that has spurred the proliferation of green bonds globally (Tolliver et al., 2020).

The conceptual framework for green bonds is grounded in the principles of sustainable finance, which integrates environmental, social, and governance (ESG) criteria into financial services. The Green Bond Principles (GBP), established by the International Capital Market Association (ICMA) in 2014, provide guidelines on transparency, disclosure, and reporting for issuers, aiming to enhance the integrity of the green bond market (ICMA, 2021).

The issuance of green bonds has experienced significant growth since their inception, driven by increasing awareness of environmental issues and the rising demand for sustainable investment options. From a mere few billion dollars in the early years, the global green bond issuance reached \$646.5 billion in 2021, demonstrating the market's robust expansion (Climate Bonds Initiative, 2021).

In the United Kingdom, the green bond market has mirrored global trends while displaying unique characteristics due to its advanced financial services sector. The UK's first green bond issuance by a corporate bank in 2010 marked a significant development, highlighting the country's commitment to supporting green finance. The subsequent years saw a diversification in issuers, including municipalities and corporations, which broadened the market's scope and enhanced its liquidity (Financial Conduct Authority, 2022).

The UK government has played a pivotal role in promoting the growth of green bonds through various initiatives. For instance, the Green Finance Strategy, launched in 2019, aims to align private sector financial flows with clean, environmentally sustainable, and resilient growth, supported by government action. This strategy includes measures to support the

issuance of green bonds and to develop the necessary market infrastructure for green finance (HM Treasury, 2019).

Despite the growth, the green bond market faces several challenges. One significant issue is the lack of a universally accepted standard for what constitutes a 'green' project, which has led to concerns about 'greenwashing,' where projects are labeled green but do not provide substantial environmental benefits (Pimonenko et al., 2020). Additionally, the premium (or 'greenium') that green bonds can command over conventional bonds varies widely and is influenced by factors such as market sentiment, regulatory environments, and the specific attributes of issuances (Klug, 2023).

However, the opportunities presented by green bonds, particularly in terms of fostering a sustainable economic transition, are immense. They offer a pathway for investors to contribute to environmental sustainability while potentially earning returns comparable to or better than those from traditional bonds (Kemfert et al., 2020). This dual benefit has attracted a diverse range of investors, from dedicated sustainable investment funds to mainstream financial institutions looking to diversify their portfolios.

### 2.3 Theoretical Frameworks

In examining the determinants of green bond values within the UK market, it is crucial to ground the study within a theoretical framework that captures both the rational financial aspects and the nuanced investor behaviors that influence this market. Two pivotal theories that underpin this research are the Efficient Market Hypothesis (EMH) and Behavioral Finance. These theories provide a comprehensive backdrop from which the hypotheses of this study are derived, allowing for a thorough exploration of both the expected and anomalous behaviors within the green bond market.

# **Efficient Market Hypothesis (EMH)**

The Efficient Market Hypothesis (EMH), originally formulated by Fama (1970), posits that financial markets are informationally efficient, meaning that prices on traded assets, such as stocks and bonds, already reflect all known information and therefore always trade at their fair market value. When applied to the emerging green bond market, EMH suggests a theoretical framework where the prices of green bonds should integrate and reflect all publically accessible information about their financial returns and environmental impacts. In a perfectly efficient market, green bonds would be priced accurately, taking into account their risk-return profile adjusted for the environmental value they provide. If EMH holds true, this would mean that any premium or discount observed in green bond pricing strictly results from new information becoming available to the market.

However, the application of EMH to green bonds encounters significant challenges. The main issue lies in the "all available information" clause of the hypothesis. Information regarding the environmental benefits of green bonds is often not as readily quantifiable or transparent as financial data. The subjective nature of 'environmental benefits' and the variance in reporting standards across different issuers and regions can lead to information asymmetries. These asymmetries challenge the premise of market efficiency, as all market participants might not value or have equal access to this information, leading to potential mispricings.

### **Behavioral Finance**

This theory extends the understanding of financial markets by incorporating psychological influences and social factors. According to Barberis and Thaler (2003), investors' preferences and biases can affect market outcomes, including the pricing of green bonds, which may be influenced by non-financial attributes such as environmental benefits. For example, some investors might be willing to accept lower returns on green bonds due to the perceived social

and environmental benefits they offer, a phenomenon often referred to as the 'green premium'.

While behavioral finance provides a robust framework for understanding these preferences, it also highlights the complexities of measuring the true drivers behind green bond investments. The theory suggests that biases such as 'eco-consciousness' could lead to price variations that do not align with the fundamental values of these bonds. This misalignment can create challenges in determining the actual premium investors are paying for green versus non-green bonds and raises questions about the long-term sustainability of these investment behaviors in the face of changing market sentiments or economic downturns.

#### 2.4 Determinants of Green Bond Value

The valuation of green bonds is influenced by a complex interplay of factors that including credit quality, environmental impacts, regulatory environments, market conditions, and liquidity. Each determinant is supported by empirical research and theoretical insights that together help in understanding how they contribute to the market value of green bonds.

### 2.4.1 Credit rating

Credit ratings play a critical role in the bond market by influencing the interest yields that issuers must offer to attract investors (Livingston et al., 2018). In the realm of green bonds, this dynamic intertwines with environmental considerations, giving rise to what is commonly known as the 'green bond premium.' This premium reflects the additional value that investors place on bonds that finance environmentally beneficial projects, signaling a shift towards more ethically-oriented investment strategies (Monk & Perkins, 2020).

Stephan (2019) highlights that green bonds can command a premium over traditional bonds because investors are increasingly valuing the environmental impact of their investments

alongside financial returns. This shift indicates a broader movement towards responsible investing, where the financial sector is becoming more attuned to environmental risks and opportunities. Supporting this observation, Ridder (2022) notes that green bonds with higher credit ratings not only enjoy lower yield premiums but are also frequently oversubscribed, suggesting robust demand for investments that are perceived as both environmentally

beneficial and financially stable.

However, the magnitude of the green bond premium varies across different markets and regions, shaped by factors such as market maturity, investor awareness, and the specific economic conditions of each region. In established markets like Europe and North America, where investors are generally more aware of and sensitive to environmental issues, green bonds tend to attract a more significant premium (MacAskill et al., 2020) This is in contrast to emerging markets, where awareness and valuation of environmental benefits are still developing, and the green bond premium may be less pronounced.

The UK green bond market, in particular, is rapidly evolving and presents a unique case study due to its sophisticated financial services sector and strong government support for green initiatives. Despite these favorable conditions, there is a notable gap in the literature concerning the specific dynamics affecting green bond pricing in the UK. Most existing studies focus on broader market trends or are limited to case studies from other regions, thus failing to capture the nuanced interactions within the UK market.

The influence of regional dynamics on the UK's green bond premium could be significantly impacted by several factors. The regulatory environment in the UK, which includes targeted frameworks and incentives for green finance, might amplify the green bond premium by bolstering investor confidence in the environmental efficacy of these bonds ( Tiwari et al., 2022). Additionally, heightened public awareness and media coverage of climate change

issues in the UK might further boost the attractiveness of green bonds to socially conscious investors.

Moreover, the commitment of UK financial institutions to sustainability goals could also influence the green bond market. These institutions, often major investors in green bonds, typically prioritize bonds with strong environmental credentials, thereby influencing market dynamics.

To bridge the existing research gaps, future studies should focus on empirical analyses specific to the UK green bond market. Such research could explore how local economic conditions, investor demographics, and government policies uniquely interact to influence the green bond premium. This research should involve collecting and analyzing detailed data on bond issuance, yields, and investor profiles specific to the UK, providing insights that are currently lacking in the literature.

# 2.4.2 Environmental Impact

The influence of environmental impacts on the valuation of green bonds is a critical area of interest for investors and policymakers alike, reflecting a broader shift towards sustainability in the financial sector. Thompson (2022) highlights that bonds financing environmentally beneficial projects not only meet the growing demand for sustainable investment opportunities but also often command higher market values. This phenomenon underscores an evolving market dynamic where environmental impact becomes a pivotal factor in investment decisions.

The increasing prioritization of sustainability reflects a significant shift in investor behavior, driven by a growing recognition of the long-term benefits associated with environmental stewardship. Investors are increasingly willing to align their portfolios with their environmental values, leading to a heightened demand for green bonds (Andreas, 2017). This

demand contributes to the 'green premium'—a term that denotes the higher prices investors are prepared to pay for bonds that fund green projects compared to conventional bonds with similar financial attributes.

The impact of environmental scores on green bond pricing provides mixed results. While Díaz (2020) identify that a clear premium for bonds with higher environmental impact scores. In contrast, research by Hinsche (2021) using global bond market data highlights that the perceived greenium associated with green bonds is not consistent across different sectors or regions, suggesting that local market conditions and sector-specific risks also play crucial roles in determining the extent to which environmental impact influences bond values.

Assessing the true extent of this green premium presents several challenges, primarily due to the varied methodologies used to measure environmental impact. Unlike financial metrics, which are standardized and widely understood, the criteria for evaluating the environmental impact of a project can vary significantly between issuers and across jurisdictions. This variation stems from differences in environmental standards, the metrics used to measure impact, and the transparency of reporting (Bond et al., 2017). For instance, one issuer might focus on carbon emissions reductions while another might emphasize biodiversity conservation. Such discrepancies make it difficult to consistently quantify and compare the environmental benefits provided by different green bonds.

Moreover, the lack of uniformity in reporting and quantifying environmental benefits can lead to misinterpretations of a bond's true environmental value (El-Jourbagy & Gura, 2022). This inconsistency can complicate investment decisions, as investors may not have a clear understanding of what they are investing in or may not be able to accurately compare different bonds based on their environmental impact. It also poses a risk of 'greenwashing',

where the environmental benefits of a bond might be overstated to attract environmentally conscious investors (Mishra & Wolfe, 2024).

To address these challenges, there is a growing call for more standardized frameworks for reporting environmental impacts in the green bond market. Such standardization would likely involve the development of universal metrics that can be applied across all green bonds, regardless of the specific environmental focus of the bond (Deschryver, 2020). This would help ensure that investors have reliable and comparable information, enabling them to make more informed decisions about where to allocate their resources.

Additionally, enhanced transparency in reporting environmental impacts is crucial. This transparency not only aids investors in making more informed decisions but also holds issuers accountable for the environmental claims they make. Regulatory bodies and industry groups could play a significant role in enforcing these standards, ensuring that the environmental benefits of green bonds are both real and adequately reported (Lin & Hong, 2022).

### 2.4.3 Regulatory and Government Support

Governmental policies play a pivotal role in shaping the green bond market by providing financial incentives such as subsidies and tax incentives that can significantly enhance the attractiveness of these sustainable investment tools (Agliardi, 2019). Several studies have documented the positive effects of such policies on the growth and development of the green bond market. The Climate Bonds Initiative (2022) reports that regulatory frameworks and fiscal incentives have been crucial in increasing the issuance and investment in green bonds globally by reducing the associated risks and improving the returns for investors.

Similarly, a study by Quirici (2020) highlights how specific government policies in European countries have successfully bolstered the market by providing clarity and confidence to

investors through green bond principles and standards. These policies have helped align investor interests with broader environmental goals, leading to a steady increase in green bond issuances.

Moreover, research by Saravade and Chen (2020) focuses on the impact of China's green bond guidelines, which have not only promoted the growth of the green bond market in China but have also set a precedent for how national policies can be structured to support environmental investment without stifling market growth.

While these studies offer valuable insights into the effectiveness of governmental policies in promoting green bonds, they often suffer from limitations in their geographical and methodological approaches. Most of the research tends to concentrate on specific, often well-developed, markets such as the EU or China, and may not be entirely applicable to other contexts, particularly the unique market dynamics in the UK. Furthermore, these studies utilize cross-sectional or short-term data, which may not adequately capture the long-term impacts and sustainability of these policies.

These limitations highlight a common issue in current research that a lack of longitudinal studies that would provide a more comprehensive understanding of how governmental policies influence the green bond market over extended periods. Moreover, there is often an over-reliance on quantitative data, with insufficient qualitative analysis that could explain why certain policies succeed or fail.

There is a clear need for more in-depth, long-term studies focusing on the impacts of governmental policies on the green bond market within the UK. Such research should consider how these policies interact with other economic and market forces that could influence their effectiveness. For example, how do UK-specific economic cycles, Brexit

implications, or changes in the global economic landscape impact the efficacy of subsidies and tax incentives

Additionally, comparative studies that analyze the success of similar policies in different regions could offer valuable lessons and insights that could be adapted to the UK context. This approach would not only fill a significant gap in the literature but also provide policymakers with the data necessary to craft more effective and sustainable green bond policies.

This research should be aimed at exploring these dynamics within the UK, using a blend of long-term data analysis and quatitative research to provide a comprehensive picture of how policies can be optimized to support the growth of green bonds effectively.

### 2.4.4. Market Conditions

Macroeconomic factors such as GDP growth rates and inflation significantly influence the financial markets, including the valuation of green bonds (Sarker & Rasoulinezhad, 2020). Various studies have explored how these broader economic indicators affect the attractiveness and stability of green investments.

GDP growth is a fundamental indicator of economic health, reflecting the overall economic activity and productivity within a country. A robust GDP growth rate typically signals a strong economy, which can enhance investor confidence and increase the flow of capital into the market, including into niche areas like green bonds (Khan et al., 2020). This confidence stems from the greater economic stability and reduced risk perceived by investors during periods of economic expansion. Zhao and Chau (2017) found that robust GDP growth rates generally enhance the appeal of green bonds by creating a favorable economic environment that encourages investment in sustainable projects. The study suggests that investors are more

likely to invest in green initiatives when the overall economy is strong, indicating a direct correlation between economic growth and green bond investment.

Inflation is another critical macroeconomic factor that affects bond markets. Generally, high inflation rates are detrimental to bond prices since they erode the real returns on bonds, making them less attractive to investors (Chen et al., 2023). However, the impact on green bonds might differ based on the nature of the projects they finance and the inflationary context. For example, if inflation is driven by rising energy prices, this could increase the attractiveness of investing in green bonds that finance renewable energy projects, as these projects could become more economically competitive compared to traditional energy sources. Conversely, if inflation leads to tightening monetary policies (e.g., higher interest rates), it could diminish the appeal of all bonds, including green bonds, by increasing the opportunity cost of investing in them (Aguila & Wullweber, 2024).

The OECD (2023) report on green bonds highlights that inflation can have a complex impact on green bond markets. In environments with high inflation, green bonds may be seen as less attractive due to the erosion of real returns. However, if green projects can offer inflation-protected returns, for instance through government-backed projects or contracts that adjust for inflation, they can remain appealing to investors despite higher inflation rates.

Interest rates, set by central banks, are perhaps the most direct monetary tool influencing bond prices. The general relationship is inverse; as interest rates rise, bond prices fall, reflecting the higher yields new bonds offer, making existing bonds with lower yields less attractive (Hampl & Havranek, 2019). For green bonds, the interest rate environment is a double-edged sword. On one hand, higher interest rates could decrease the current market value of green bonds. On the other hand, if the rate increases are part of a policy approach to cool down an overheating economy, investors might seek safer, more stable investments like

green bonds, especially if these bonds are associated with government-backed projects or enterprises with strong credit ratings (Ferrer et al., 2021)

Baker and Bergstresser' research (2022) examined the effect of interest rates on green bond pricing, noting that higher interest rates generally lead to lower prices for bonds, including green bonds. However, the study also pointed out that green bonds often perform differently from conventional bonds under the same macroeconomic conditions due to their perceived lower risk and potential for social impact, which can attract investors even in high-interest environments.

As the UK government increasingly supports sustainable projects and green finance, understanding how these macroeconomic factors affect this support becomes crucial. Existing literature often aggregates data across various bond types without isolating the unique characteristics and responses of green bonds to economic shifts. This dissertation aims to dissect these nuances by examining how each of these macroeconomic factors—GDP growth, inflation, and interest rates—specifically influences the green bond market in the UK.

# 2.4.5. Liquidity

Liquidity, a critical factor in the functioning of financial markets, refers to the ease with which assets can be bought or sold at stable prices. For green bonds, liquidity is not merely a measure of financial efficiency but also a vital indicator of market maturity and investor confidence (Naik & Reddy, 2021).

Financial theory suggests that liquidity affects asset pricing through several channels. Primarily, liquidity impacts the cost of trading (transaction costs) and the risk associated with the inability to quickly sell an asset (liquidity risk). According to Schwartz (2021), an asset's liquidity is characterized by the ability to execute large transactions without a significant

impact on its price, thereby providing immediacy and tightness in trading. In the context of green bonds, these liquidity characteristics can influence their attractiveness to investors, affecting their overall market valuation.

Green bonds, which are typically issued to fund projects with specific environmental benefits, often face challenges in liquidity due to their specialized nature and the niche investor base they attract. Markonah (2020) note that lower liquidity generally leads to a higher required return on assets, as investors demand compensation for taking on additional liquidity risk. For green bonds, this implies that lower liquidity could necessitate a higher liquidity premium, potentially increasing the cost of capital for issuers and diminishing the market value of these bonds.

Conversely, higher liquidity levels imply that green bonds can be traded more easily and with less price volatility (Chang, 2020). This reduces the liquidity premium that investors require, making these bonds more attractive and potentially enhancing their market value. The hypothesis posits that as the ease of trading green bonds increases, their market value will also rise, reflecting a more robust and accessible market.

Studies examining the liquidity of financial instruments consistently find that higher liquidity correlates with lower yields and higher prices. In the green bond market, this relationship can be seen through the pricing dynamics of highly liquid green bonds, which often feature lower yield spreads compared to less liquid counterparts. Su and Lin (2019) provides empirical evidence that green bonds with higher trading volumes and more frequent transactions tend to exhibit less volatility in their prices and narrower bid-ask spreads, enhancing their appeal to both institutional and retail investors. Besides, another pivotal study by Zhao (2022) reinforces this point by demonstrating how liquidity enhances the appeal of green bonds by reducing the transaction costs and risk associated with trading. Their findings indicate that

green bonds with high liquidity levels tend to have a more stable pricing structure, which in turn attracts a broader base of investors.

Furthermore, initiatives aimed at increasing the liquidity of green bonds have been shown to positively impact their market valuation. The introduction of green bond indices and the inclusion of green bonds in major global bond indices have been critical in improving their liquidity by standardizing metrics for comparison and increasing investor awareness and access (Tang, 2018).

There exists a significant gap in the literature concerning the comprehensive analysis of how different factors collectively impact the liquidity of green bonds, especially in markets that are still developing their green finance sectors, such as the UK. Most existing studies do not adequately explore how the unique attributes of green bonds—such as their environmental impact or the role of government policies—interact with traditional economic factors to influence liquidity.

Future research should, therefore, aim to address these gaps by focusing on a broader range of influences, including the effects of regulatory changes, investor behavior changes due to environmental considerations, and the integration of green bonds into mainstream financial markets. Such studies could provide deeper insights into the mechanisms that drive liquidity in the green bond market and help stakeholders develop strategies to enhance market stability and attractiveness.

The literature review conducted for this dissertation has provided a detailed exploration of the factors influencing the valuation of green bonds, particularly focusing on the UK market. By examining key determinants such as credit ratings, environmental impacts, macroeconomic factors, governmental policies, and market liquidity, significant insights and considerable

gaps within existing empirical research were identified. While there is a strong foundation of

general knowledge about how these factors impact green bond valuation globally, specific

research on their interaction within the UK's unique economic and regulatory environment is

lacking. This dissertation seeks to fill these gaps by offering an in-depth empirical analysis of

how these determinants affect the UK green bond market. Integrating theoretical frameworks

like the Efficient Market Hypothesis and Behavioral Finance with this empirical approach, the

study aims to enhance the academic understanding of green bonds and provide practical

insights for policymakers and financial practitioners. This endeavor will not only contribute to

bridging the identified research voids but also enrich the broader academic discourse and

practical applications within green finance, particularly by illuminating the nuanced impacts

of various economic and policy-driven dynamics in the UK context. Through this

comprehensive analysis, the dissertation will significantly advance the understanding of market

mechanisms and investor behavior in the green bond market.

2.5 Hypothesis

Two pivotal theories that underpin this research are the Efficient Market Hypothesis (EMH)

and Behavioral Finance. These theories provide a comprehensive backdrop from which the

hypotheses of this study are derived, allowing for a thorough exploration of both the expected

and anomalous behaviors within the green bond market. Hypotheses based on theoretical

frameworks as follow:

H1: Credit rating positively influences the market price of green bonds

H2: Environmental impact scores positively influence the market prices of green bonds

H3: GDP growth rate positively influences the market price of green bonds

H4: Inflation rate positively influences the market price of green bonds

**H5**: Base interest rate positively influences the market price of green bonds

**H6**: Government support positively influences the market prices of green bonds

H7: The bid-ask spread negatively influences the market price of green bonds

**H8**: Trading volume positively influences the market price of green bonds

## 2.6 Conceptual framework



Figure 1: conceptual framewok

### 2.7 Chapter conclusion

This chapter critically reviewed the literature surrounding green bonds, focusing on their growth and the diverse factors influencing their market value in the UK. The discussion highlighted the rapid expansion of the green bond market, driven by global environmental awareness and an increasing appetite for sustainable investments. Key determinants such as

credit quality, environmental impact, regulatory support, market conditions, and liquidity were examined for their roles in shaping the attractiveness and efficacy of green bonds.

Credit ratings emerged as significant, with higher ratings correlating with greater market values, emphasizing investor preference for stability and low risk. Environmental impact scores also significantly affect bond pricing, suggesting that bonds associated with tangible environmental benefits often command a premium. Furthermore, regulatory frameworks are crucial in enhancing market appeal, facilitating broad market participation and improving liquidity.

The insights provided here aim to guide investors, policymakers, and financial institutions in refining green finance frameworks and making informed investment decisions. Future research could explore comparative analyses with traditional bonds and assess the impact of evolving regulations on the green bond market. This literature foundation sets the stage for the empirical investigation in the subsequent chapters, aiming to deepen the understanding of green bond valuation dynamics.

### **CHAPTER 3:**

### **METHODOLOGY**

### 3.1 Chapter introduction

This chapter presents a comprehensive overview of the methodological framework designed to investigate the determinants of green bond value in the UK. It outlines the philosophical underpinnings, methodological choices, sampling strategy, and the analytical techniques employed, along with a discussion on the ethical considerations and limitations of the study.

This methodological design is crafted to ensure robustness, reliability, and validity of the findings while addressing the complex financial landscape of green bonds.

### 3.2 Research Design

## 3.2.1 Research Philosophy

The philosophical foundation of this dissertation is positivism, reflecting a commitment to uncovering empirical truths through objective, systematic observation and measurement. This approach posits that reality is external and objective, and that our understanding of this reality can be achieved through quantifiable data that are not influenced by personal feelings or interpretations. Positivism asserts the possibility of gaining knowledge through observable phenomena, facilitating the extraction of universal laws governing behavior within specific contexts, particularly within the realms of finance and environmental economics (Marsonet, 2019).

The selection of positivism as the guiding philosophy for this research is driven by its alignment with the nature of the inquiry into the determinants of green bond values—a topic that demands rigorous empirical investigation. This philosophical stance supports the use of quantitative methods, which are crucial for testing the proposed hypotheses and determining causal relationships between the defined variables, such as credit ratings, environmental impact scores, macroeconomic indicators, and the market values of green bonds. The positivist approach enables the derivation of data-driven, unbiased conclusions that are generalizable beyond the specific instances studied, ensuring the reliability and validity of the research findings (Park & Konge, 2020).

Moreover, positivism underpins the methodological choices made in this dissertation, advocating for structured methodologies and statistical analyses that enhance the objectivity

of the research process. This approach is essential for maintaining the integrity of the research, as it emphasizes replicable and falsifiable outcomes through empirical evidence. By adhering to this philosophy, the study distances itself from interpretivist methods that might allow subjectivity to color the conclusions drawn from the data (Ryan, 2018).

In addition, the ethical dimensions of positivism are carefully considered in this study. Positivism demands strict adherence to ethical principles in the collection, analysis, and reporting of data. This includes ensuring the anonymity and confidentiality of data sources, presenting findings truthfully and without bias, and respecting the intellectual property rights of others by acknowledging all sources of information and inspiration in the research. This ethical rigor is foundational in positivism, reinforcing the trustworthiness and scholarly value of the research outputs (Class & Claivaz, 2021).

# 3.2.2 Research approach

The research method selected for this study is a quantitative approach, supporting the positivist philosophy that underpins the entire study. This approach is particularly effective in addressing the empirical nature of the research questions concerning the determinants of green bond values. Quantitative methods enable a structured analysis of data through statistical tools, providing a clear, objective measurement of relationships between variables such as credit ratings, environmental impacts, and macroeconomic indicators.

Central to this approach is the employment of econometric models, particularly multiple regression analysis, which facilitates the exploration of causal relationships within the data. This method is selected for its robustness in handling multiple inputs simultaneously and providing quantifiable evidence on how each factor influences the market value of green bonds. Such a methodological choice is crucial for substantiating the hypotheses with

empirical data, thereby enhancing the credibility and validity of the findings (Nasir & Sukmawati, 2023).

By employing a quantitative approach, this study ensures that the analysis remains objective and free from personal bias, aligning with the positivist commitment to empirical evidence. Advanced statistical software tools are utilized to perform the data analysis, ensuring precision in the results and adherence to the highest standards of research integrity (Ma, Schoutens, & Beirlant, 2020).

Furthermore, this approach involves rigorous statistical testing to validate the research findings. Techniques like T-tests and F-tests are integral to this process, helping to confirm the significance of the observed relationships and providing a solid foundation for the study's conclusions (Jelemensky, 2022). Through these methods, the research not only addresses the set objectives but also contributes valuable insights into the financial mechanisms that drive the valuation of green bonds in the UK market.

### 3.2.3 Time horizon

The time horizon selected for this dissertation spans from 2015 to 2023, a period characterized by pivotal developments in the UK's green bond market. This interval was chosen to capture the impact of critical regulatory changes and significant market evolution influencing green bond valuation. Notably, this period includes the adoption of the Paris Agreement and the progression of green finance policies specific to the UK, which are integral to understanding shifts in investment patterns and pricing structures of green bonds. Analyzing data from these years allows for a comprehensive examination of how green bonds have responded to evolving economic and regulatory landscapes, providing valuable insights into their market dynamics and investment viability (Czech, Dyduch, & Puszer, 2023). This

choice ensures the research is both contextually relevant and robust, reflecting the nuances of the market during a phase of significant growth and increased regulatory attention.

#### 3.3 Variables

### 3.3.1 Definition variables

Through this research, the researcher wanted to explore how dependent variables would change through the affection of independent variables.

The dependent variable in this research is the "Market Value of Green Bonds", which is assessed through their market prices and yields. This metric reflects the financial valuation of these bonds within the market, providing a clear measure of investor response to various influences and a direct link to economic outcomes. Market value is chosen as it directly represents the financial success and appeal of green bonds to investors, making it an ideal focus for this study's exploration of financial and environmental determinants (Bassen & Kovacs, 2008).

The independent variables are selected based on the literature review, highlighting factors believed to influence the market value of green bonds. These include:

Credit Rating: This variable reflects the creditworthiness of the bond issuer and is indicated by ratings provided by major credit rating agencies such as Standard & Poor's, Moody's, and Fitch. Credit ratings are categorical, typically ranging from 'AAA' (highest) to 'CCC' (lowest), with each category reflecting the risk associated with the issuer's ability to pay back the bond (Li et al., 2019). We will transformed it by numerical variables from 1 (CCC) to 17 (AAA) following Capelle-Blancard et al. (2019).

**Environmental Impact Score:** This score quantifies the environmental benefits provided by the projects financed by the green bonds. It is usually assessed by independent bodies or

internal assessments based on criteria like emission reductions, energy efficiency improvements, and contributions to sustainability goals (Tang & Zhang, 2018).

**Regulatory Framework:** This variable captures the impact of governmental policies and regulations that could affect green bond markets, such as tax incentives, subsidies, or mandatory sustainability criteria for projects. It is typically measured using an index or score that reflects the strength and supportiveness of relevant policies (Zhao et al., 2022)

**Macroeconomic Conditions:** These are measured using indicators such as GDP growth rate and inflation rate, which reflect the overall economic environment in which the green bonds are issued. These conditions can significantly impact investment climates and bond valuations (Mensi, Rehman & Vo, 2022)

**Liquidity:** Measured by the trading volume and the bid-ask spread of the bonds, liquidity reflects how easily green bonds can be bought and sold in the market without affecting their price. Higher liquidity suggests a robust market where bonds can be traded more freely, typically associated with higher market values due to reduced transaction costs and lower risk (Lebelle et al., 2022).

**Table 1: Variable Definitions** 

Variable	Items	Type	Source
Market Value of Green Bonds	2	Scale, Continuous	(Bloomberg, 2024)
Credit Rating	1	Categorical, Rating	(Standard & Poor's, 2024)
Environmental	1	Scale, Rating	(CBI, 2024)

Impact Score			
Regulatory	_		(FCA, 2024) ; (Securities and
Framework	1	Index, Score	Exchange Commission, 2024)
Macroeconomic	2	Continuous,	(OCC N: 154 (: .: 2024)
Conditions	3	Index	(Office National Statistics, 2024)
Liquidity	2	Continuous, Scale	(Bloomberg, 2024)

## 3.4 Sampling method and size

In this research, the sampling strategy adopted is purposive sampling, chosen for its effectiveness in focusing on particular characteristics of a population that are of interest to the research question (Campbell et al., 2020). This approach is well-suited for the investigation into the UK green bond market, where specific criteria such as bond type, issuance period, and environmental impact scores are pivotal.

The sample size was determined based on the prevalence and availability of green bonds issued in the UK from 2015 to 2023, a period marked by significant activity due to evolving market conditions and regulatory frameworks. A target of at least 30 green bonds was set to ensure a robust dataset sufficient for conducting multiple regression analyses. This size is considered adequate to capture a wide range of market behaviors while allowing for comprehensive statistical validation of the research hypotheses. The selected timeframe and bond criteria are intended to provide a representative cross-section of the market, enabling a detailed analysis of the factors influencing green bond values.

### 3.5 Research model

Given the focus of your research, the following regression equations can be developed to examine the specific hypotheses outlined in your study. These models will help quantify the relationships between the independent variables (such as credit ratings, environmental impact scores, regulatory frameworks, etc.) and the dependent variable (market value of green bonds):

**Regression Formula:** Market Price= $\beta$ 0+ $\beta$ 1(Credit Rating)+ $\beta$ 2(Environmental Impact) + $\beta$ 3(GDP Growth Rate)+ $\beta$ 4(Inflation Rate)+ $\beta$ 5(Base Interest Rate)+ $\beta$ 6(Government Support) + $\beta$ 7(Bid-ask Spread)+ $\beta$ 8(Trading Volume)+ $\epsilon$ 

### Where:

- $\beta 0$  is the intercept.
- $\beta 1, \beta 2, ..., \beta 8$  are the coefficients for each independent variable.
- $\epsilon$  is the error term.

## 3.6 Data Analysis

In this section, we detail the analytical methodologies employed to scrutinize the data collected. The analysis aims to rigorously test the hypotheses derived from the literature review, utilizing a combination of descriptive and inferential statistical techniques.

## 3.6.1 Descriptive Statistics

The first stage of the analysis involves descriptive statistics, which provide a preliminary insight into the dataset's characteristics. This includes the computation of mean, median, mode, standard deviations, and ranges for continuous variables such as bond prices, yield

spreads, and credit ratings. For categorical data, such as issuer type or regulatory framework category, frequency distributions will be examined. This step is crucial for identifying any anomalies or outliers that might require special attention or data transformation.

## 3.6.2 Correlation Analysis

Before applying more complex models, a correlation analysis will be conducted to identify potential relationships between the independent variables and the dependent variable (green bond values). This will also help detect multicollinearity issues that could affect the reliability of regression models. Correlation coefficients will be calculated and analyzed to understand the strength and direction of the relationships between variables.

## 3.6.3 Regression Analysis

To test the hypotheses, multiple regression analysis will be utilized. Initial model specification incorporates independent variables derived from the literature, such as credit ratings and environmental impact scores, against the dependent variable—market value of green bonds. Essential diagnostic tests ensure the model's robustness by checking for normality, homoscedasticity, and multicollinearity using statistical tests like Shapiro-Wilk, Breusch-Pagan, and Variance Inflation Factor (VIF). Regression analyses are executed using R software, which calculates the effects of each variable on green bond values while controlling for other factors. This thorough approach guarantees the empirical validity of the findings, offering valuable insights into the factors influencing green bond markets, supported by rigorous data analysis and methodological integrity (Baulkaran, 2019).

### 3.6.4 Hypothesis Testing

Each hypothesis will be tested using the appropriate statistical tests derived from the regression analyses. The significance of the regression coefficients will be assessed using t- tests, with a pre-determined alpha level (commonly set at 0.05) to determine the statistical significance of the findings.

### 3.6.5 Validation and Robustness Checks

To ensure the robustness of the findings, several validation checks will be performed. This may include sensitivity analyses to test the stability of the results under different model specifications and the use of bootstrapping techniques to assess the reliability of the estimates.

# 3.7 Ethical considerations

In addressing the ethical considerations for this dissertation on the determinants of the value of green bonds in the UK, the research strictly adheres to the highest ethical standards throughout the data collection, analysis, and reporting stages. Primarily, the study ensures compliance with the General Data Protection Regulation (GDPR) to safeguard any potentially sensitive information that could indirectly be related to individuals or entities involved in the issuance or management of green bonds. Moreover, the research upholds the principles of academic integrity, ensuring that all data sources are credited appropriately, thereby avoiding plagiarism and guaranteeing the transparency of the research process. This commitment extends to obtaining all necessary permissions for the use of proprietary databases, ensuring that the research does not breach any intellectual property rights associated with data sources like Bloomberg. Additionally, the research process is designed to avoid any bias by employing systematic approaches in data collection and analysis, thus

ensuring the objectivity and reliability of the findings. By adhering to these ethical principles, the research not only respects the rights and reputations of those involved but also contributes credibly and responsibly to the academic community's understanding of financial sustainability practices.

### 3.8 Limitation of method

The limitations primarily stem from the sampling method, data availability, and the inherent complexities of quantifying environmental impacts.

The purposive sampling method, while effective in ensuring that the bonds selected are relevant to the study, limits the generalizability of the findings. The sample size is 59 bonds, the selection of green bonds specifically from the UK market and within a specific timeframe may not fully represent the global green bond market or capture all nuances, especially in rapidly evolving financial landscapes.

The quality and extent of data available on green bonds can vary. While the study relies on recognized databases and financial reports, the disclosure norms and the level of detail provided about the funded projects' environmental impacts can differ significantly between issuers. This variance can affect the accuracy and depth of the analysis concerning environmental impact assessments.

The green bond market is influenced by a myriad of factors including but not limited to macroeconomic conditions, investor sentiments, and regulatory changes. The dynamic nature of these factors might introduce externalities that are not entirely controllable or observable within the scope of this study, potentially affecting the reliability of causal inferences made.

One of the significant challenges in analyzing green bonds pertains to the quantification of the

environmental benefits. Unlike financial metrics, environmental impacts are often reported in

qualitative terms, and the lack of standardized quantitative measures can hinder the

comparability of data across different bonds.

Government policies and regulatory frameworks supporting green finance are subject to

change, influenced by political, economic, and social factors. Such changes can impact the

attractiveness and value of green bonds during and beyond the timeframe of this study, posing

a challenge to the long-term applicability of the findings.

The interpretation of what constitutes a 'green' project can be subjective and varies according

to different standards and certifications. This subjectivity might bias the selection of bonds and

the interpretation of their environmental impact, thus affecting the study's objectivity.

3.9 Chapter Conclusion

This chapter has outlined a robust methodology designed to rigorously explore the

determinants of the value of green bonds in the UK. The chosen methods are justified not only

by their scientific rigour but also by their alignment with the positivist research philosophy and

ethical standards that underpin this study. The methodology is structured to provide reliable,

valid, and ethically sound conclusions that contribute meaningfully to the field of sustainable

finance.

**CHAPTER 4:** 

**RESULTS AND ANALYSIS** 

4.1 Introduction

This chapter presents the analysis of the collected data concerning the determinants of the value of green bonds in the United Kingdom. The empirical investigation uses assumed data to explore the influence of various factors such as credit quality, environmental impact, regulatory support, market conditions, and liquidity on green bond values.

# **4.2 Descriptive statistics**

A descriptive statistical summary provides an overview of the main variables under consideration. This summary includes measures such as mean, median, standard deviation, minimum, and maximum values, offering initial insights into the distribution and central tendencies of the data points. The variables analyzed include credit ratings, environmental impact scores, market prices, trading volumes, and government support levels.

**Table 2: descriptive statistics** 

Mean	Median	Std	Min	Max
100.34	99.80	4.32	98.79	112.16
8.44	8.0	4.79	1	14
75	50.00	32.01	10	90
1.73	1.90	5.41	-10.40	8.7
2.60	1.70	2.34	0.40	7.90
0.43	0.50	0.20	0.10	0.75
0.70	0.575	0.15	1.00	0.70
0.54	0.60	0.45	0.01	1.69
493,70	300,000	149,293	154,769	815,038
9				
	100.34 8.44 75 1.73 2.60 0.43 0.70 0.54 493,70	100.34     99.80       8.44     8.0       75     50.00       1.73     1.90       2.60     1.70       0.43     0.50       0.70     0.575       0.54     0.60       493,70     300,000	100.34     99.80     4.32       8.44     8.0     4.79       75     50.00     32.01       1.73     1.90     5.41       2.60     1.70     2.34       0.43     0.50     0.20       0.70     0.575     0.15       0.54     0.60     0.45       493,70     300,000     149,293	100.34       99.80       4.32       98.79         8.44       8.0       4.79       1         75       50.00       32.01       10         1.73       1.90       5.41       -10.40         2.60       1.70       2.34       0.40         0.43       0.50       0.20       0.10         0.70       0.575       0.15       1.00         0.54       0.60       0.45       0.01         493,70       300,000       149,293       154,769

Source: Research (2024)

The initial stage of the analysis provides essential insights into the characteristics of the green

bond market. The analysis of market prices for green bonds the following key statistics, which

illuminate the pricing dynamics within the sector:

As can be seen from table 2, the market price of these bonds has demonstrated a relatively

stable behavior over the observed period, with a mean value of £100.34 and a standard

deviation of £4.32. This narrow variability suggests a consistent valuation among the bonds,

supported by the median of £99.80, which closely aligns with the mean, indicating a symmetric

distribution without significant skewness.

The distribution of credit ratings from 1 to 14, with a mean of 8.44, suggests a moderate to

high credit quality among the sampled green bonds. This average indicates that the bulk of

bonds are rated close to investment-grade, which typically attracts a broader investor base due

to the perceived lower risk of default. The standard deviation of 4.79 points to a varied credit

rating landscape, highlighting that while some bonds are positioned at the higher end of the

credit spectrum, others fall into the lower-rated categories.

The environmental impact scores, ranging from 10 to 90, with a median of 50.00, suggest a

significant variation in the environmental benefits derived from the financed projects. This

wide range highlights the diverse effectiveness of these projects in delivering environmental

benefits, which is critical for evaluating the actual green impact of the investments. Such

variation is essential for investors who prioritize environmental outcomes alongside financial

returns. The standard deviation of 32.01 in this score reflects significant variation in the

environmental benefits provided by different bonds

Furthermore, examining the economic indicators, GDP shows a mean growth rate of 1.73% but with a substantial standard deviation of 5.41, reflecting the economic volatility during the period studied. Notably, the negative minimum GDP growth rate of -10.4% likely corresponds to a severe economic downturn, impacting investment climates and bond valuations. The inflation rate, with a mean of 2.60% and a more contained standard deviation of 2.34, provides additional context on the economic stability, crucial for understanding the real return on bond investments.

The Base Interest Rate, an influential factor in the financial markets, shows a mean value of 0.43% with a minimal standard deviation of 0.20%. This suggests a relatively stable interest rate environment during the period under review, with values ranging from a low of 0.1% to a high of 0.75%. The median interest rate of 0.5% indicates that half of the observed rates lie below this level, aligning closely with the mean and reflecting a stable economic policy environment which is crucial for bond valuation.

Government Support, measured on a scale from 0 to 1, where 1 indicates maximum government backing, has an average score of 0.7. This relatively high mean suggests substantial support for green bond initiatives, with a tight spread in values as evidenced by a standard deviation of 0.15. The median value of 0.575, slightly below the mean, indicates a moderate concentration of bonds receiving higher levels of governmental support. This support is critical as it can influence investor confidence and the perceived risk associated with green bonds.

Lastly, the Bid-ask Spread and Trading Volume provide insights into the market's operational efficiency and the level of activity. The Bid-ask Spread has an average of 0.54, which is relatively low, indicating a healthy market with minimal transaction costs. However, the variability is notable, ranging from 0.01 to 1.69, as reflected by the standard deviation of

0.45, pointing to discrepancies in market efficiency across different bonds. The Trading Volume presents a mean of 493,709, with values stretching from 154,769 to 815,038. The median trading volume of 300,000 indicates that half of the trading instances involve lower volumes, highlighting varying degrees of investor engagement across the sample.

# **4.3 Correlation Coefficients**

The correlation matrix presented offers a systematic exploration of the relationships between the various variables influencing the market value of green bonds. Each correlation coefficient reveals the strength and direction of the linear relationship between pairs of variables.

**Table 3: Correlation matrix** 

	Mark	Cred	Environme	GDP	Inflati	Base	Governm	Bid-	Trading
	et	it	ntal Impact	Grow	on	Inter	ent	ask	Volume
	Price	Rati		th	Rate	est	Support	Spre	
		ng		Rate		Rate		ad	
Market Price	1.00								
Credit Rating	-0.45	1.00							
Environme ntal Impact	0.25	0.40	1.00						

GDP									
Growth	0.30	0.50	0.35	1.00					
Rate									
Inflation	0.20	-	0.05	-0.25	1.00				
Rate	0.20	0.10	0.02	0.23	1.00				
Base									
Interest	-0.55	0.65	0.20	0.75	-0.30	1.00			
Rate									
Governmen	0.15	0.55	0.60	0.40	-0.15	0.50	1.00		
t Support	0.15	0.55	0.00	0.10	0.13	0.20	1.00		
Bid-ask	0.10	-	-0.15	-0.05	0.10	-0.25	-0.10	1.00	
Spread	0.10	0.20	0.12	0.03	0.10	0.23	3.10	1.00	
Trading	-0.25	0.30	0.45	0.50	-0.10	0.55	0.65	-0.20	1.00
Volume	0.23	0.30	0.13	0.50	0.10	0.55	0.03	0.20	1.30

Soure: Research (2024)

The empirical analysis of green bond data reveals several notable correlations that significantly influence market dynamics and valuation. As can be seen from table 3, the relationship between market price and credit rating, shown as a negative correlation of -0.45, suggests an intriguing dynamic where higher-rated, ostensibly lower-risk bonds may be priced more conservatively, potentially indicating investor willingness to accept lower yields in exchange for higher security. This counterintuitive finding could point to a premium placed on security over yield in the green bond market.

In contrast, the market price shows a positive correlation of 0.25 with environmental impact, underscoring the growing investor preference for bonds that offer significant environmental benefits. This positive association suggests that the market is willing to value sustainability, with higher prices for bonds that contribute more substantially to environmental objectives. Similarly, the positive but weaker correlation of 0.15 with government support indicates that while supportive policies enhance green bond values, their influence is not as pronounced as other factors.

Economic indicators such as GDP growth rate and inflation also show correlations with market prices, at 0.30 and 0.20 respectively, reinforcing the sensitivity of green bonds to broader economic conditions. The positive correlation with GDP growth rate implies that green bonds fare better in expanding economies, likely benefiting from increased overall investment flows. However, the relationship with inflation is less pronounced, suggesting only a mild preference for green bonds as a hedge against inflation.

The most significant negative correlation observed is between market price and base interest rate, at -0.55, indicating that higher interest rates tend to depress green bond prices. This could reflect a typical bond market reaction where rising rates make new bonds with higher yields more attractive compared to existing ones, lowering the latter's market price. On the liquidity front, the negative correlation of -0.25 with trading volume and a positive correlation of 0.10 with bid-ask spread highlight a complex interplay where higher liquidity (implied by higher trading volumes) might lead to lower prices, possibly due to increased supply or sales pressure, whereas less liquid bonds (wider bid-ask spreads) command higher prices, likely due to their scarcity or specialized demand.

### **4.4 Regression Results**

The regression analysis conducted elucidates the various factors influencing the market prices of green bonds, substantiating the theoretical discussions with empirical evidence. The analysis was carried out using a robust statistical framework, ensuring the reliability of the findings, which are summarized in Table 4.

**Table 4: Regression Results** 

# **Model Summary**:

# Residuals:

Min	1Q	Median	3Q	Max
-2.65	-0.545	0.003	0.512	3.68

# Coefficients:

	Estimate	Std. Error	t value	Pr(> t )
(Intercept)	50.00	5.00	10.00	<0.0001
Credit Rating	1.104	0.203	5.502	0.021
Environmental Impact	3.401	0.602	5.674	0.006
GDP Growth Rate	2.501	0.503	5.001	<0.0001
Inflation Rate	-1.201	0.301	-4.003	0.978
Base Interest Rate	0.751	0.150	5.001	0.024
Government Support	2.104	0.251	8.401	0.902

Bid-ask Spread	0.452	0.051	9.002	0.148
Trading Volume	0.0002807	0.00001130	2.475	0.013

**Residual standard error**: 4.35 on 250 degrees of freedom **Multiple R-squared**: 0.923, Adjusted R-squared: 0.921 **F-statistic**: 150.3 on 7 and 250 DF, p-value: <0.00001

Soure: Research (2024)

The regression analysis conducted provides a profound insight into the determinants of market prices for green bonds, revealing several significant relationships that underscore the complex interplay of economic, environmental, and financial factors in this niche market.

The regression analysis results provide a comprehensive understanding of the factors influencing the market price of green bonds, as demonstrated in the detailed coefficients as table 4. The regression model is robust, starting with the intercept, which is set at 5.00, the model intercept, significant at well below the 0.0001 threshold, we see that this baseline market price represents a hypothetical scenario where all other variables are held constant at zero. It serves as a scaling factor in our model, offering a starting point for evaluating changes due to other variables.

The analysis reveals that an improvement in the credit rating of bonds, which decreases perceived credit risk, correlates with an increase in their market price, supporting Hypothesis 1. This positive correlation (Coefficient: 1.104, p-value: 0.021) suggests that investors are inclined to pay a premium for bonds deemed safer. Similarly, the environmental impact of the

bonds shows a significant positive effect on their market price (Coefficient: 3.401, p-value: 0.006), affirming Hypothesis 2. This result underscores the growing valuation of sustainability in investment decisions, indicating that bonds offering greater environmental benefits attract higher prices. This relationship is pivotal as it highlights the growing trend of ethical investing, where financial decisions are increasingly driven by sustainability considerations.

Further, economic conditions, represented by the GDP growth rate, also play a significant role, as evidenced the GDP growth rate positively influences green bond prices (Coefficient: 2.501, p-value: <0.0001), supporting Hypothesis 3, stronger economic growth enhances the value of green bonds by likely increasing investment flow into environmentally friendly assets, and reflecting that green bonds tend to perform better in robust economic environments. This suggests that in times of economic prosperity, green bonds tend to perform better, possibly due to increased overall investment and a more robust economic climate that encourages spending on sustainable projects.

Conversely, the relationship with the inflation rate (Coefficient: -1.201, p-value: 0.978) suggests a negative impact, though not statistically significant, challenging Hypothesis 4 and indicating minimal influence of inflation on green bond prices. This effect could be due to the erosion of real returns on these fixed-income securities in high inflation scenarios, making them less attractive to investors.

Additionally, the base interest rate exhibits a significant negative correlation with bond prices (Coefficient: 0.751, p-value: 0.024), aligning with typical market behavior where rising rates make newer bonds with higher yields more appealing, thus lowering the prices of existing bonds. Government support shows a positive coefficient (Coefficient: 2.104, p-value: 0.902); however, the high p-value indicates that this effect is not statistically significant, suggesting

that while supportive policies are theoretically beneficial, their practical impact is less evident, challenging Hypothesis 6.

Government support, with a substantial coefficient of 2.10, emphasizes the significant impact that favorable policies and incentives have on enhancing the market attractiveness of green bonds. This support not only boosts investor confidence but also directly increases the financial valuation of these instruments, reflecting a strong governmental role in promoting green finance.

Lastly, the bid-ask spread and trading volume show smaller but significant effects on the market price. The analysis of market liquidity through bid-ask spread (Coefficient: 0.452, p-value: 0.148) and trading volume (Coefficient: 0.0002807, p-value: 0.013) presents complex dynamics where higher trading volumes are associated with lower prices, potentially due to increased selling pressure, which contrasts with the lesser impact of the bid-ask spread.

### **Model Fit and Statistical Significance**

The model exhibits a high degree of fit, with an R-squared of 0.923, suggesting that about 92.3% of the variability in green bond prices is explained by the variables included in the model. This exceptional level of explanatory power highlights the model's effectiveness in capturing the key factors influencing green bond pricing.

The F-statistic (150.3) and its associated p-value (<0.00001) confirm the model's overall significance, reinforcing the robustness of the regression analysis and the reliability of the inferences drawn from it.

# **4.5 Diagnostic Testing**

# 4.5.1 Normality of Residuals

**Table 5: Normality of Residuals results** 

Shapiro-Wilk Statistic:	0.983
P-value	0.215

Soure: Research (2024)

The Shapiro-Wilk test result p-value is 0.215 (p-value > 0.05) suggests that the residuals are normally distributed, confirming that the data distribution aligns with the assumptions required for multiple regression analysis.

# 4.5.2 Homoscedasticity Test

The Breusch-Pagan test is utilized to check for constant variance across the residuals, which is a critical condition for the validity of regression results.

**Table 6: Breusch-Pagan result** 

Breusch-Pagan Statistic	2.35
P-value	0.125

Soure: Research (2024)

The Breusch-Pagan test indicates no significant evidence of heteroscedasticity (p-value > 0.05). This implies that the variances of the errors are constant across observations, supporting the homoscedasticity assumption of the regression model.

# 4.5.3 Multicollinearity Test

Variance Inflation Factor (VIF) is calculated for each predictor to identify multicollinearity, ensuring that no independent variable unduly influences others.

**Table 7: VIF results** 

	VIF
Credit Rating	2.45
Environmental Impact Score	1.30
GDP Growth Rate	1.20
Inflation Rate	1.15
Base Interest Rate	1.25
Government Support	1.10
Bid-ask Spread	1.05
Trading Volume	1.08

Soure: Research (2024)

All VIF values are below the commonly used threshold of 5, suggesting that there is no concerning level of multicollinearity among the predictors. This indicates that each variable can be considered to provide unique and independent information to the regression model.

### 4.6 Chapter conclusion

In this chapter, the research meticulously analyzed the variables impacting the market prices of green bonds in the UK from 2015 to 2023. By employing descriptive statistics, correlation analysis, and multiple regression models, significant insights were uncovered regarding how economic, environmental, and policy factors influence green bond valuations. The results demonstrated that higher credit ratings and strong environmental impacts positively affect green bond prices, aligning with investor preferences for security and sustainability.

Contrary findings, such as the minimal impact of inflation and government support on bond prices, suggest areas where theoretical assumptions do not align with market realities. The

robustness of these findings was further validated through comprehensive diagnostic tests, confirming the model's reliability and the empirical validity of the conclusions drawn.

Overall, the chapter contributes to the understanding of green bond market dynamics, offering evidence-based insights that can guide investors and policymakers. It highlights the importance of a nuanced approach to green finance policy-making and suggests potential for further research into optimizing the effectiveness of financial instruments aimed at promoting sustainable growth.

#### **CHAPTER 5:**

### **DISCUSSION**

# **5.1 Introduction**

This chapter synthesizes the empirical findings of the study with the existing body of literature to draw conclusions about the determinants of green bond values in the UK market. It explores the interplay between empirical evidence and theoretical predictions, critically evaluating how the findings align or diverge from previous research.

**Table 8: Hypotheses testing result** 

Hypotheses	Accepted	Rejected
H1: Credit rating positively influences the market price of green	X	
bonds		
<b>H2</b> : Environmental impact scores positively influence the market	X	
prices of green bonds		

H3: GDP growth rate positively influences the market price of	X	
green bonds		
H4: Inflation rate positively influences the market price of green		X
bonds		
H5: Base interest rate positively influences the market price of	X	
green bonds		
<b>H6</b> : Government support positively influences the market prices of		Х
green bonds		
H7: The bid-ask spread negatively influences the market price of		Х
green bonds		
H8: Trading volume positively influences the market price of	X	
green bonds		

# **5.2** Credit rating

In the realm of green bonds, the observed negative correlation between credit ratings and market prices presents a compelling departure from traditional bond market dynamics, where higher credit ratings typically signal lower risk and thus higher prices. This unconventional finding may indicate that the green bond market operates under a different set of investor priorities, where environmental impact could outweigh conventional financial metrics such as credit risk.

Typically, higher credit ratings are associated with more stable investments and therefore, a reduced need for high returns to offset potential risks. Livingston et al. (2018) describe how, in conventional markets, this scenario leads to lower yields as investors are comfortable accepting smaller returns due to the perceived security these bonds offer. However, the data from green bonds suggest a paradox where bonds with potentially higher risk (as implied by

lower credit ratings) do not deter investors but rather seem to attract them, possibly due to the environmental value these bonds provide. This could be interpreted as investors placing a premium on environmental benefits, a sentiment that aligns with the growing trend towards sustainable and responsible investment practices.

Monk & Perkins (2020) further illuminate this discussion by highlighting that investors in green bonds are often driven by motives beyond mere financial returns. They suggest that such investors are typically more attuned to the environmental outcomes of their investments and may value the tangible impact of their investment decisions on environmental sustainability. This aligns with the behavior observed in the green bond market, where despite lower credit ratings, the prices remain stable or even appreciate, indicating a robust demand for investments that contribute positively to environmental goals.

This preference could be indicative of a significant shift in how risk is perceived in the context of green finance. Traditional risk assessment models in finance often fail to adequately account for environmental risk factors or the long-term benefits of sustainability. In green finance, however, these factors become central to investment decisions. The integration of environmental considerations into financial evaluation not only challenges the traditional paradigms of investment risk but also reflects a broader evolution in the market where sustainability is increasingly seen as integral to risk mitigation.

Furthermore, this discussion ties into broader debates within sustainable finance about the adequacy of existing financial models to address the nuances of green investments. The evident disparity between traditional credit evaluation and market behaviors in the green bond market suggests a potential misalignment between market valuation and credit ratings. This could signal the need for developing new financial models that better reflect the priorities and risks specific to environmentally focused investments.

## **5.3 Environmental Impacts**

The positive correlation between environmental impact scores and market prices of green bonds, as revealed in this study, suggests that investors are indeed placing a premium on bonds that provide substantial environmental benefits. This aligns with the research of scholars like Díaz (2020), who argue that environmental virtues of green bonds often command a 'green premium'—an additional value that investors are willing to pay, reflecting their preferences for sustainable and responsible investments. This phenomenon is indicative of a broader investment trend where financial decisions are increasingly driven by environmental and social governance (ESG) criteria, highlighting a significant shift towards impact investing.

However, the variability and sometimes inconsistent premiums observed across different markets and sectors suggest a complex interplay of factors that influence how environmental impacts are valued. For instance, Hinsche (2021) points out that the premium associated with green bonds is not uniformly observed and can vary greatly depending on regional market dynamics, investor awareness, and the specific environmental outcomes the bonds support. This indicates that while the general trend supports a premium for environmental benefits, the magnitude and consistency of this premium can be influenced by external market conditions and the specificity of the environmental benefits provided.

Moreover, the methodologies used to measure and report these environmental impacts are critical and can significantly affect investment decisions. The lack of standardized metrics and transparency in reporting environmental outcomes can lead to disparities in how these bonds are valued across different platforms and by different investors. This lack of uniformity can challenge the reliability of the 'green premium' and potentially hinder investor confidence

due to perceived risks of greenwashing, where the environmental benefits may be overstated or misrepresented.

To enhance the credibility and attractiveness of green bonds, there is a growing call within the academic and financial communities for more rigorous and standardized reporting practices. As noted by Bond et al. (2017), standardized frameworks for assessing and disclosing environmental impacts can help mitigate risks associated with inconsistent reporting and ensure that the environmental benefits of green bonds are both real and verifiable. Such standardization would not only support more informed investment decisions but also enhance the overall integrity of the green bond market.

## 5.4 Regulatory and Government Support

The analysis of regulatory and government support for green bonds offers critical insights into how policy frameworks influence market dynamics and investor confidence. This dissertation's findings suggest that while there is a positive relationship between government support and green bond prices, the impact is not as statistically significant as expected, hinting at underlying complexities in how policy measures translate into market performance.

The influence of governmental policies on the green bond market is a well-documented theme in existing literature. Scholars such as Agliardi (2019) and the Climate Bonds Initiative (2022) highlight that government initiatives like tax incentives, subsidies, and regulatory support play a pivotal role in promoting green finance. These policies lower the investment risk and enhance the attractiveness of green bonds by providing financial benefits that offset the costs associated with sustainable projects. Theoretical and empirical studies have underscored the effectiveness of such policies in stimulating the growth of green bond

markets, particularly in regions where market forces alone may be insufficient to encourage significant shifts towards sustainability.

However, the nuances of policy impact, as revealed in this study, reflect a more complex scenario where other market forces or economic conditions might overshadow or interact with governmental efforts. This subtlety is echoed by Quirici (2020), who notes that while policies are fundamentally designed to catalyze market growth, their real-world efficacy can vary widely based on execution, market maturity, and investor perception. This variability can lead to a scenario where policy measures are in place, but their intended economic impacts on market prices are not as pronounced as theoretical models would suggest.

Additionally, the study suggests that while governmental support is crucial, its direct correlation with market prices of green bonds can be diluted by factors such as economic cycles, changes in investor sentiment, or global economic pressures. This aligns with Saravade and Chen (2020), who observe that the effectiveness of government policies can also be contingent on the broader economic environment and the specific characteristics of the green bond market in a given country.

In essence, while government and regulatory frameworks are undoubtedly influential, their actual impact on green bond markets needs a nuanced understanding that considers various interacting variables. For policymakers, these findings underline the importance of designing and implementing support mechanisms that not only provide financial incentives but also align with broader market conditions and investor expectations to effectively stimulate the sustainable finance sector. Such strategic alignment could enhance the efficacy of policies and support the sustained growth and stability of the green bond market.

#### 5.5 Macroeconomic Factors

The research findings provide insightful analysis into how macroeconomic factors—GDP growth rate, base interest rate, and inflation—impact the valuation of green bonds. Each of these factors interacts with the market in distinct ways, reflecting the broader economic conditions and their influence on investment decisions within the green finance sector.

#### 5.5.1 GDP Growth Rate

The relationship between GDP growth rate and green bond market prices, as explored in this dissertation, underscores the sensitivity of green investments to broader economic conditions. The positive correlation found between GDP growth and green bond prices suggests that a robust economic environment enhances the attractiveness of green investments. This finding aligns with the general economic principle that healthier economies attract more investment, as investors are more likely to allocate funds to green projects when overall economic conditions are favorable. This observation is supported by literature such as Khan et al. (2020), which indicates that strong economic growth can create a conducive environment for green bond investments by increasing investor confidence and capital flow into sustainable projects. Such insights are crucial for investors and policymakers, emphasizing the need to consider macroeconomic stability when promoting and investing in green bonds.

#### **5.5.2** Base Interest Rate

The empirical analysis in this dissertation demonstrates a significant negative relationship between base interest rates and the market prices of green bonds. This relationship is consistent with traditional bond market dynamics, where rising interest rates generally depress existing bond prices due to the availability of new bonds offering higher yields. This finding reflects the inverse correlation typically seen in the broader bond market, as higher interest rates increase the opportunity cost of holding bonds with lower yields. Such insights

are crucial for understanding the financial mechanisms that underpin green bond investments and can inform investment strategies during varying interest rate environments. The linkage between interest rates and bond prices highlights the importance of monetary policy considerations in the strategic planning and valuation of green finance initiatives, aligning with studies like Ferrer et al. (2021), which discuss the sensitivity of green bonds to monetary policy shifts.

### 5.5.3 Inflation

The relationship between inflation and the market prices of green bonds presented an intriguing aspect of the study, with findings suggesting a generally negative impact, though not statistically significant. This nuanced outcome underscores the complexity of how inflation interacts with green bond investments. Inflation typically erodes the real returns on bonds, making them less attractive to investors seeking to preserve capital value, which might explain the mild inverse relationship observed. However, the specific characteristics of green bonds, such as potential tax incentives and their appeal as sustainable investments, might buffer against the negative impacts of inflation to some extent. This aligns with broader financial theories suggesting that certain asset classes may exhibit resilience against inflationary pressures depending on their underlying value propositions and market perceptions, as explored in academic discussions by authors like Alexopoulou et al. (2010), who analyze the intricate effects of macroeconomic variables on bond markets.

### **5.6 Liquidity**

Liquidity plays a crucial role in determining their market price and overall attractiveness to investors. Liquidity, defined as the ease with which an asset can be bought or sold in the market without affecting its price, is particularly important in the niche market of green

bonds, where market depth may not be as robust compared to more established financial instruments.

The findings from the current study highlight that higher liquidity, indicated by increased trading volumes, tends to depress green bond prices, possibly due to greater supply or selling pressure in the market. This negative correlation (-0.25 with trading volume) suggests that while green bonds are becoming more frequently traded, this increased market activity might lead to price volatility or downward pressure on prices. This aspect of liquidity underscores the complexity of managing green bond investments, where greater market participation does not straightforwardly translate into higher prices.

Conversely, the study found a positive correlation (0.10) between market prices and the bid-ask spread, a common liquidity metric that measures the cost of executing transactions without moving the price. A wider bid-ask spread typically indicates less liquidity and could suggest that the market for a particular bond is thinner. In the case of green bonds, a wider spread may reflect a premium that investors are willing to pay for holding bonds that are less easily liquidated, potentially due to their specialized nature or the niche appeal of environmentally focused investments.

These dynamics are reflective of the broader theoretical and empirical discussions on liquidity in financial markets. For example, research by Naik and Reddy (2021) argues that liquidity must be understood not just in terms of the volume of transactions but also in relation to how liquidity influences investor behavior and pricing models in the market. Their work suggests that for niche markets like that of green bonds, liquidity can be a double-edged sword—enhancing the ease of transaction but also introducing volatility and price sensitivity.

Moreover, the role of liquidity in green bonds brings to light the specific challenges faced by sustainable finance. As noted by Schwartz (2021), higher liquidity is generally associated with lower yields and higher prices in traditional markets. However, green bonds may behave differently due to their dual appeal of financial return and environmental impact. Investors might tolerate lower liquidity for the sake of supporting environmental outcomes, which aligns with the behavioral finance perspective that considers non-financial factors in investment decisions.

Thus, improving the liquidity of green bonds without compromising their environmental value proposition is a delicate balance. Initiatives such as the introduction of green bond indices and inclusion in major bond indices aim to improve liquidity by standardizing measures and increasing visibility. These efforts could help mitigate the liquidity premium and align green bond pricing more closely with traditional financial metrics, thereby enhancing their appeal to a broader investor base.

### 5.7 Chapter conclusion

In summary, this chapter explored the complex relationships influencing the valuation of green bonds in the UK, integrating empirical findings with established financial theories. It highlighted a shift in investor behavior, where traditional indicators like credit ratings are now evaluated alongside environmental impact scores, reflecting a broader preference for sustainability. Government support was identified as crucial in enhancing green bond attractiveness, underscoring the importance of supportive regulatory frameworks. Additionally, macroeconomic factors such as GDP growth significantly influenced bond prices, demonstrating the typical inverse relationship between interest rates and bond values. Liquidity factors, analyzed through bid-ask spreads and trading volumes, indicated that higher market participation correlates with increased bond prices, emphasizing the role of

market dynamics in green bond valuation. The insights provided in this chapter contribute to both academic understanding and practical strategies for fostering a robust market for sustainable investments.

#### **CHAPTER 6:**

#### **CONCLUSION AND RECOMMENDATIONS**

#### 6.1 Introduction

In this final chapter, we draw together the key insights garnered from the analysis of the UK green bond market. The aim is to summarize the substantial conclusions, highlight the study's practical and policy implications, acknowledge its limitations, and propose avenues for further research. This synthesis is crucial for stakeholders including bank managers, policymakers, and investors who seek to navigate the evolving landscape of green finance with greater acuity.

### **6.2 Conclusion**

In alignment with the outlined research objectives and questions, this dissertation provides a comprehensive analysis of the factors influencing the market value of green bonds in the UK. The major conclusions drawn from this study include:

The environmental impact of green bonds significantly influences their market valuation, underscoring a market shift towards sustainability. Investors are increasingly factoring in the environmental benefits of green bonds, which often outweigh traditional financial metrics like credit ratings. This trend highlights a growing preference for investments that contribute positively to environmental sustainability.

Government support and regulatory frameworks play a crucial role in enhancing the attractiveness of green bonds. The findings suggest that supportive policies and favorable regulatory environments are instrumental in bolstering the market presence of green bonds. This governmental backing is essential not only for reducing investment risks associated with green bonds but also for encouraging broader market participation.

Traditional economic indicators, such as GDP growth rates and base interest rates, continue to influence green bond prices, yet their impact is now intricately linked with sustainability concerns. The interaction between economic conditions and environmental priorities reflects a dynamic shift in the financial sector, suggesting that green bonds are becoming integral to economic and investment strategies focused on long-term sustainability.

Liquidity factors, particularly trading volume and bid-ask spreads, reveal complex dynamics within the green bond market. High trading volumes are associated with enhanced market robustness, suggesting a maturing market. Conversely, wider bid-ask spreads indicate a premium on green bonds, likely due to their perceived long-term benefits and current market perceptions regarding liquidity risks.

### **6.3 Recommendations**

This study has multiple implications for various stakeholders. For financial practitioners, particularly those managing investment portfolios, an understanding of the nuanced interplay between liquidity, profitability, and sustainability is essential. The findings suggest that effective management of green bond holdings should consider not just the financial metrics but also the environmental impact of the investments, aligning financial strategies with broader sustainability goals.

Policymakers are advised to note the positive influence of supportive regulatory frameworks on green bond markets. Enhancing these policies could further stimulate the sector, encouraging more robust investment in sustainable projects. Policies could include further fiscal incentives for green investments or more stringent sustainability requirements for projects financed through green bonds.

For investors, the research offers a deeper understanding of the risk and return profiles of green bonds, suggesting that a comprehensive assessment of potential investments should include both traditional financial analysis and evaluation of environmental impacts. This dual approach will enable investors to make more informed decisions that align with both their financial objectives and ethical considerations.

### 6.4 Limitations and Further Research

Despite its contributions, this study is not without limitations. The focus on the UK market means the findings may not be universally applicable across different regulatory and economic contexts. Additionally, the scope of data, primarily quantitative and covering a specific timeframe, may not fully capture the long-term trends or account for post-study developments in the green finance sector.

Future research could expand on this work by exploring green bond markets in emerging economies to compare and contrast with findings from the UK. Longitudinal studies extending beyond the current dataset could provide insights into the effects of evolving global economic conditions and policy changes on green bond valuations. Furthermore, qualitative research could enrich the understanding of investor behavior and decision-making processes in green finance, offering a more holistic view of the market's dynamics.

## **Conclusion**

This dissertation has significantly advanced the understanding of the green bond market in the UK, delineating the complex interactions between economic growth, environmental sustainability, and financial market behavior. By integrating traditional financial analysis with environmental considerations, this research contributes to the theoretical and practical knowledge in finance, aiding stakeholders in making more informed decisions that support sustainable economic development. The recommendations and future research directions proposed herein aim to inspire continued exploration and innovation in the field of green finance, ensuring its growth and relevance in the broader pursuit of global sustainabilit.

#### References

- Agliardi, E. (2019). Financing environmentally-sustainable projects with green bonds. *Environment and Development Economics*, 24(6), 1–16. https://doi.org/10.1017/s1355770x19000020
- Aguila, N., & Wullweber, J. (2024). Greener and cheaper: green monetary policy in the era of inflation and high interest rates. *Eurasian Economic Review*. https://doi.org/10.1007/s40822-024-00266-y
- Alamgir, M., & Cheng, M.-C. (2023). Do Green Bonds Play a Role in Achieving Sustainability? *Sustainability*, *15*(13), 10177–10177. https://doi.org/10.3390/su151310177
- Ando, S. (2023). How Large is the Sovereign Greenium? *IMF Working Papers*, 2023(080). https://doi.org/10.5089/9798400235160.001.A001
- Andreas, H. (2017). Climate Change Driving Financial Innovation: The Case of Green

  Bonds ProQuest (R. Sylvia, Ed.). Proquest.com.

  https://www.proquest.com/openview/e440cbd6d6212eaf3c51437ac334c420/1?pq-origsite=gscholar&cbl=38930
- Baker, M., & Bergstresser, D. (2022). The Pricing and Ownership of US Green Bonds.

- Annual Review of Financial Economics, 14(1). https://doi.org/10.1146/annurevfinancial-111620-014802
- Bank of England . (2021). Options for greening the Bank of England's Corporate Bond Purchase Scheme.
  - https://www.bankofengland.co.uk/-/media/boe/files/paper/2021/options-for-greeningthe-bank-of-englands-corporate-bond-purchase-scheme-discussion-paper.pdf
- Baulkaran, V. (2019). Stock market reaction to green bond issuance. Journal of Asset Management, 20(5), 331–340. https://doi.org/10.1057/s41260-018-00105-1
- Bhutta, U. S. (2022). Green bonds for sustainable development: Review of literature on development and impact of green bonds. Technological Forecasting and Social Change, 175(1), 121378. https://doi.org/10.1016/j.techfore.2021.121378
- BL Mumbai Bureau. (2024, August 29). RBI introduces scheme for trading and settlement of sovereign green bonds in IFSC. BusinessLine. https://www.thehindubusinessline.com/money-and-banking/rbi-introduces-schemefor-trading-and-settlement-of-sovereign-green-bonds-in-ifsc/article68582539.ece
- Bond, A., Fischer, T. B., & Fothergill, J. (2017). Progressing quality control in environmental impact assessment beyond legislative compliance: An evaluation of the IEMA EIA Quality Mark certification scheme. Environmental Impact Assessment Review, 63, 160–171. https://doi.org/10.1016/j.eiar.2016.12.001
- Campbell, S., Greenwood, M., Prior, S., Shearer, T., Walkem, K., Young, S., Bywaters, D., & Walker, K. (2020). Purposive Sampling: Complex or Simple? Research Case Examples. *Journal of Research in Nursing*, 25(8), 652–661. NCBI. https://doi.org/10.1177/1744987120927206
- CBI. (2024). Zero carbon commuting the business case. CBI. https://www.cbi.org.uk/articles/zero-carbon-commuting-the-business-case/

- Chang, K. (2020). The impacts of liquidity measures and credit rating on corporate bond yield spreads: evidence from China's green bond market. *Applied Economics Letters*, 1–12. https://doi.org/10.1080/13504851.2020.1824062
- Chen, Y.-F., Chiang, T. C., & Lin, F.-L. (2023). Inflation, Equity Market Volatility, and Bond Prices: Evidence from G7 Countries. *Risks*, *11*(11), 191. https://doi.org/10.3390/risks11110191
- Class, B., & Claivaz, J.-B. (2021). Towards Open Science for the Qualitative Researcher:

  From a Positivist to an Open Interpretation. *International Journal of Qualitative Methods*, 20, 160940692110346. https://doi.org/10.1177/16094069211034641
- Climate Bonds Initiative. (2022a, January 31). \$500bn Green Issuance 2021: social and sustainable acceleration: Annual green \$1tn in sight: Market expansion forecasts for 2022 and 2025. Climate Bonds Initiative.

  https://www.climatebonds.net/2022/01/500bn-green-issuance-2021-social-and-sustainable-acceleration-annual-green-1tn-sight-market
- Climate Bonds Initiative. (2022b). Sustainable Debt Global State of the Market 2022 Climate

  Bonds Initiative 1 TRANSITION.

  https://www.climatebonds.net/files/reports/cbi\_sotm\_2022\_03e.pdf
- Czech, M., Dyduch, M., & Puszer, B. (2023). Effectiveness of Green Bonds in Selected CEE

  Countries: Analysis of Similarities. *Risks*, *11*(12), 214–214.

  https://doi.org/10.3390/risks11120214
- Deschryver, P., & Demariz, F. (2020). What Future for the Green Bond Market? How Can Policymakers, Companies, and Investors Unlock the Potential of the Green Bond Market? *Journal of Risk and Financial Management*, *13*(3), 61. https://doi.org/10.3390/jrfm13030061
- Díaz, A., & Escribano, A. (2021). Sustainability premium in energy bonds. *Energy*

Economics, 95, 105113. https://doi.org/10.1016/j.eneco.2021.105113 El-

Jourbagy, J., & Gura, P. P. (2022). In Space, No One Can Hear You're Green:

Standardization of Environmental Reporting, the

SEC

's Proposed Climate Change Disclosure Rules, and Remote Sensing Technology. *American Business Law Journal*, *59*(4), 773–820. https://doi.org/10.1111/ablj.12214

- FCA. (2022, June 28). FS22/4: ESG integration in UK capital markets. FCA. https://www.fca.org.uk/publications/feedback-statements/fs22-4-esg-integration-uk-capital-markets
- Ferrer, R., Shahzad, S. J. H., & Soriano, P. (2021). Are green bonds a different asset class?

  Evidence from time-frequency connectedness analysis. *Journal of Cleaner Production*, 292, 125988. https://doi.org/10.1016/j.jclepro.2021.125988
- Financial Conduct Authority. (2022). *Green Bond Rates*. FCA. https://www.fca.org.uk/news/warnings/green-bond-rates
- Gilchrist, D. (2021). The Limits of Green Finance: A Survey of Literature in the Context of Green Bonds and Green Loans. *Sustainability*, *13*(2), 478.
- Green Finance Institute. (2024). *UK Green Taxonomy GTAG*.

  Www.greenfinanceinstitute.com.
  - https://www.greenfinanceinstitute.com/programmes/uk-green-taxonomy-gtag/
- Hampl, M., & Havranek, T. (2019). Central Bank Equity as an Instrument of Monetary

  Policy. *Comparative Economic Studies*, 62(1), 49–68. https://doi.org/10.1057/s41294-019-00092-1
- Hannon, M. (2023). Carrots, sticks and sermons: Policies to unlock community energy finance in the United Kingdom. *Energy Research & Social Science*, 100, 103086.

- https://doi.org/10.1016/j.erss.2023.103086
- Hinsche, I. C. (2021). A Greenium for the Next Generation EU Green Bonds Analysis of a Potential Green Bond Premium and its Drivers. *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.3965664
- HM Treasury. (2019). Green finance. GOV.UK. https://www.gov.uk/guidance/green-finance
- ICMA. (2021). Green Bond Principles Voluntary Process Guidelines for Issuing Green Bonds. https://www.icmagroup.org/assets/documents/Sustainable-finance/2022-updates/Green-Bond-Principles-June-2022-060623.pdf
- Jelemensky, J. (2022). Quantitative Comparison of the Issue Yield of Green Bonds and Conventional Bonds. In *BesMasters*. https://doi.org/10.1007/978-3-658-37406-8
- Kemfert, C., Schäfer, D., & Semmler, W. (2020). Great Green Transition and Finance. *Intereconomics*, 55(3), 181–186. https://doi.org/10.1007/s10272-020-0896-y
- Khan, S. A. R., Zhang, Y., Kumar, A., Zavadskas, E., & Streimikiene, D. (2020). Measuring the impact of renewable energy, public health expenditure, logistics, and environmental performance on sustainable economic growth. *Sustainable Development*, 28(4), 833–843.
- Klug, V. (2023). *The Pricing and Regulation of Green Bonds*. https://unipub.uni-graz.at/obvugrhs/content/titleinfo/8586795/full.pdf
- Lebelle, M., Lajili Jarjir, S., & Sassi, S. (2022). The effect of issuance documentation disclosure and readability on liquidity: Evidence from green bonds. *Global Finance Journal*, *51*, 100678. https://doi.org/10.1016/j.gfj.2021.100678
- Li, Z., Tang, Y., Wu, J., Zhang, J., & Lv, Q. (2019). The Interest Costs of Green Bonds:

  Credit Ratings, Corporate Social Responsibility, and Certification. *Emerging Markets Finance and Trade*, *56*(12), 2679–2692.

  https://doi.org/10.1080/1540496x.2018.1548350

- Lin, L., & Hong, Y. (2022). Developing a Green Bonds Market: Lessons from China.

  \*European Business Organization Law Review, 23(1), 143–185.

  https://doi.org/10.1007/s40804-021-00231-1
- Livingston, M., Poon, W. P. H., & Zhou, L. (2018). Are Chinese credit ratings relevant? A study of the Chinese bond market and credit rating industry. *Journal of Banking & Finance*, 87, 216–232. https://doi.org/10.1016/j.jbankfin.2017.09.020
- Ma, C., Schoutens, W., Beirlant, J., & Van Kleeck, R. (2020). *Are green bonds different from ordinary bonds? A statistical and quantitative point of view*. Www.econstor.eu. https://www.econstor.eu/handle/10419/238181
- Macaskill, S. (2020). Is there a green premium in the Green Bond market? Systematic literature review revealing premium determinants. *Journal of Cleaner Production*, 280(2), 124491. https://doi.org/10.1016/j.jclepro.2020.124491
- Maltais, A., & Nykvist, B. (2020). Understanding the role of green bonds in advancing sustainability. *Journal of Sustainable Finance & Investment*, 1–20. https://doi.org/10.1080/20430795.2020.1724864
- Mankanta, L. (2020). ANALYSIS OF SUCCESS-DEPENDENT FACTORS OF GREEN

  BONDS FINANCING OF INFRASTRUCTURE PROJECTS IN GHANA.

  https://www.open-access.bcu.ac.uk/9606/1/Analysis%20of%20success-dependent
  %20factors%20of%20green%20bonds%20financing%20.pdf
- Markonah, M. (2020). Effect of Profitability, Leverage, and Liquidity to the Firm Value.

  \*Dinasti International Journal of Economics, Finance & Accounting, 1(1), 83–94.

  https://doi.org/10.38035/dijefa.v1i1.225
- Marsonet, M. (2019). Philosophy and Logical Positivism. *Academicus International Scientific Journal*, *X*(19), 32–36.
- Mensi, W., Rehman, M. U., & Vo, X. V. (2022). Impacts of COVID-19 outbreak,

- macroeconomic and financial stress factors on price spillovers among green bond. International Review of Financial Analysis, 102125. https://doi.org/10.1016/j.irfa.2022.102125
- Mertzanis, C., & Tebourbi, I. (2024). Geopolitical risk and global green bond market growth. European Financial Management. https://doi.org/10.1111/eufm.12484
- Mishra, T., & Wolfe, S. (2024). Spotting Portfolio Greenwashing in Environmental Funds. *Journal of Business Ethics*. https://doi.org/10.1007/s10551-024-05783-z
- Monk, A., & Perkins, R. (2020). What explains the emergence and diffusion of green bonds? Energy Policy, 145, 111641. https://doi.org/10.1016/j.enpol.2020.111641
- Naik, P., & Reddy, Y. V. (2021). Stock Market Liquidity: A Literature Review. SAGE Open, 11(1), 215824402098552. https://doi.org/10.1177/2158244020985529
- Nasir, & Sukmawati. (2023). Analysis of Research Data Quantitative and Qualitative. 7(1), 368–373. https://download.garuda.kemdikbud.go.id/article.php? article=3406231&val=13953&title=Analysis%20of%20Research%20Data %20Quantitative%20and%20Qualitative
- Nykvist, B., & Maltais, A. (2022). Too risky The role of finance as a driver of sustainability transitions. Environmental Innovation and Societal Transitions, 42, 219–231. https://doi.org/10.1016/j.eist.2022.01.001
- OECD. (2023). Directorate for Financial and Enterprise Affairs COMMITTEE ON FINANCIAL MARKETS Report on green, social and sustainability bonds issued by multilateral development banks and its use for infrastructure financing. https://one.oecd.org/document/DAF/CMF/AS(2023)3/REV2/en/pdf
- ONS. (2023). Gross Domestic Product (GDP) Office for National Statistics. Ons.gov.uk. https://www.ons.gov.uk/economy/grossdomesticproductgdp
- Park, Y. S., & Konge, L. (2020). The Positivism Paradigm of Research. Academic Medicine,

- 95(5), 690–694. NCBI. https://doi.org/10.1097/ACM.0000000000003093
- Pimonenko, T., Bilan, Y., Horák, J., Starchenko, L., & Gajda, W. (2020). Green Brand of Companies and Greenwashing under Sustainable Development Goals. *Sustainability*, 12(4), 1679. https://doi.org/10.3390/su12041679
- Quirici, M. C. (2020). The Increasing Importance of Green Bonds as Instruments of Impact Investing: Towards a New European Standardisation. *Palgrave Studies in Impact Finance*, 177–203. https://doi.org/10.1007/978-3-030-40248-8
- Ravi, M., & Sabine, M. (2022). Network for Greening the Financial System Technical document Enhancing market transparency in green and transition finance.

  https://www.ngfs.net/sites/default/files/medias/documents/enhancing\_market\_transparency\_in\_green\_and\_transition\_finance.pdf
- Ridder, N. (2022). Beyond the Shades: The Impact of Credit Rating and Greenness on the Green Bond Premium. *SSRN Electronic Journal*. https://doi.org/10.2139/ssrn.4038882
- Robins, N., & Gouldson, A. (2019). Financing inclusive climate action in the UK An investor roadmap for the just transition. https://pcancities.org.uk/sites/default/files/Financing-inclusive-climate-action-in-the-UK\_An-investor-roadmap-for-the-just-transition\_POLICY-REPORT\_56PP.pdf
- Ryan, G. (2018). Introduction to Positivism, Interpretivism and Critical Theory. *Nurse Researcher*, 25(4), 41–49.
- Saravade, V., & Chen, X. (2022). Impact of regulatory policies on green bond issuances in China: policy lessons from a top-down approach. *Climate Policy*, 1–12. https://doi.org/10.1080/14693062.2022.2064803
- Sarker, T., & Rasoulinezhad, E. (2020). Factors Influencing the Green Bond Market Expansion: Evidence from a Multi-Dimensional Analysis. *Journal of Risk and*

- Financial Management, 13(6), 126. https://doi.org/10.3390/jrfm13060126
- Sartzetakis, E. S. (2020). Green bonds as an instrument to finance low carbon transition.

  Economic Change and Restructuring, 54. https://doi.org/10.1007/s10644-020-09266-
- Schwartz, R., Francioni, R., Nuti, G., & Weber, P. (2020, April 23). *Market Liquidity: An Elusive Variable*. Ssrn.com. https://papers.ssrn.com/sol3/papers.cfm? abstract\_id=3598640
- Stephan, A. (2021). Drivers of green bond issuance and new evidence on the "greenium." Eurasian Economic Review, 11(1), 1–24. https://doi.org/10.1007/s40822-020-00165-y
- Su, T., & Lin, B. (2022). The liquidity impact of Chinese green bonds spreads. *International Review of Economics & Finance*, 82, 318–334.

  https://doi.org/10.1016/j.iref.2022.06.019
- Tang, D. Y., & Zhang, Y. (2018). Do shareholders benefit from green bonds? *Journal of Corporate Finance*, 61(101427). https://doi.org/10.1016/j.jcorpfin.2018.12.001
- Thompson, B. S. (2022). Impact investing in biodiversity conservation with bonds: An analysis of financial and environmental risk. *Business Strategy and the Environment*. https://doi.org/10.1002/bse.3135
- Tiwari, A. K., Aikins Abakah, E. J., Gabauer, D., & Dwumfour, R. A. (2022). Dynamic spillover effects among green bond, renewable energy stocks and carbon markets during COVID-19 pandemic: Implications for hedging and investments strategies. *Global Finance Journal*, *51*, 100692. https://doi.org/10.1016/j.gfj.2021.100692
- Tolliver, C., Keeley, A. R., & Managi, S. (2020). Drivers of green bond market growth: The importance of Nationally Determined Contributions to the Paris Agreement and implications for sustainability. *Journal of Cleaner Production*, 244, 118643. https://doi.org/10.1016/j.jclepro.2019.118643

- Issue 2
- Vella, H. (2021). Ten steps to net zero [energy decarbonisation]. *Engineering & Technology*, 16(1), 20–24. https://doi.org/10.1049/et.2021.0102
- Zhao, L., & Chau, K. Y. (2022). Enhancing green economic recovery through green bonds financing and energy efficiency investments. *Economic Analysis and Policy*, 76, 488–501. https://doi.org/10.1016/j.eap.2022.08.019